

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Shtaltovna, Anastasiya

### Working Paper Knowledge gaps and rural development in Tajikistan: Agricultural advisory services as a panacea?

ZEF Working Paper Series, No. 117

**Provided in Cooperation with:** Zentrum für Entwicklungsforschung / Center for Development Research (ZEF), University of Bonn

*Suggested Citation:* Shtaltovna, Anastasiya (2013) : Knowledge gaps and rural development in Tajikistan: Agricultural advisory services as a panacea?, ZEF Working Paper Series, No. 117, University of Bonn, Center for Development Research (ZEF), Bonn

This Version is available at: https://hdl.handle.net/10419/88303

### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU

# 



Zentrum für Entwicklungsforschung Center for Development Research University of Bonn

# Working Paper 117

Anastasiya Shtaltovna

Knowledge gaps and rural development in Tajikistan: Agricultural advisory services as a panacea?



ZEF Working Paper Series, ISSN 1864-6638 Department of Political and Cultural Change Center for Development Research, University of Bonn Editors: Joachim von Braun, Manfred Denich, Solvay Gerke, Anna-Katharina Hornidge

### Authors' address

Dr. Anastasiya Shtaltovna Senior Researcher Centre for Development Research Department of Political & Cultural Change (ZEFa) Walter-Flex. Str. 3, 53113 Bonn, Germany Staff page: http://www.zef.de/staff/922.html mail to: shtaltov@uni-bonn.de Phone: +49 (0)228-73-49-17

www.zef.de

# Knowledge gaps and rural development in Tajikistan.

# Agricultural advisory services as a panacea?

Anastasiya Shtaltovna

# CONTENTS

CONTENTS1			
1.	ABSTRACT	. 2	
2.	INTRODUCTION	.3	
3.	OVERVIEW OF AGRICULTURAL ADVISORY SERVICE MODELS	.5	
4.	RESEARCH METHODOLOGY	.6	
5.	THE RURAL ECONOMY OF TAJIKISTAN	.7	
6.	SOURCES OF KNOWLEDGE FOR FARMERS IN TAJIKISTAN	.9	
7.	DISCUSSION: HOW SHOULD AGRICULTURAL ADVISORY SERVICES LOOK LIKE IN TAJIKISTAN?	17	
REFERENCES			

## **1 ABSTRACT**

In this paper I look at knowledge systems and the channels of innovation diffusion in Tajikistan. In particular I look at the formation of the agricultural advisory services as a vital source of knowledge and innovation for farmers during the transition process. After the end of the Soviet Union and the civil war in Tajikistan, the knowledge available to farmers can be described as a mix of, on the one hand, traces and fragments originating from the Soviet agricultural and educational system represented by the universities, research institutes and academy of sciences; and on the other hand, western-style knowledge, mainly introduced by development agencies. The paper analyses the functioning of and interaction between the international donors, NGOs, Tajik government and academic institutions. Donors are relatively new actors in the field of knowledge dissemination, but important ones. In this paper I show how under the framework of 'development', i.e. the rhetoric, organization and infrastructure of development, different donors play their own games, some of them geo-political. At the same time they provide support for the functioning of local NGOs and help them to sustain; also donors are used by the Tajik political actors for their own purposes. From this discussion I derive suggestions on how agricultural advisory services could be organized in Tajikistan, working on some local weaknesses and building on existing assets, traditions and networks, reflecting the interplay of the main actors and local needs.

**Keywords:** agricultural advisory services, extension, knowledge, innovation, rural development, development cooperation, Tajikistan, Central Asia

### **2 INTRODUCTION**

### Swan, Pike And Crawfish

When partners can't agree Their dealings come to naught And trouble is their labour's only fruit.

Once Crawfish, Swan and Pike Set out to pull a loaded cart, And all together settled in the traces; They pulled with all their might, but still the cart refused to budge! The load it seemed was not too much for them: Yet Crawfish scrambled backwards, Swan strained up skywards, Pike pulled toward the sea. Who's guilty here and who is right is not for us to say-But anyway the cart's still there today.

Ivan Andreevich Krylov (1769-1844 / Russia), English translation Victor Zinger

In this paper I aim to assess local epistemic cultures, actors and structures of agriculturally oriented knowledge and innovation development and the local channels of innovation diffusion<sup>1</sup>. Agricultural knowledge systems in Tajikistan undergo changes as part of the transformation process (Boboyorov 2013). During the Soviet era, knowledge for agriculture was produced together with agricultural universities, colleges, research institutes, the Ministry of Agriculture, and other actors involved in agricultural production. Within the bigger network of education and research institutions of the former Soviet Union, these institutions were linked with each other (Morgounov and Zuidema 2001). The results were targeted at and transferred to *kolkhozes and sovkhozes* that used to be the main production units of collectivized agriculture.

After the breakdown of the Soviet Union and a number of reforms in agriculture, new agricultural producers in Tajikistan have emerged. The following farm types can be identified: (1) household plots, (2) individual and family *dekhan* farms, (3) collective *dekhan* farms and (4) agricultural enterprises (more details in Section 5). The system of knowledge production and sharing has changed through the transformation process as well. The earlier mentioned Soviet network of education and research in Tajikistan

<sup>&</sup>lt;sup>1</sup> The author would like to acknowledge the Federal Ministry of Education and Research (BMBF) of Germany for the financial support provided to the the research proejct "Epistemic Cultures and Innovation Diffusion in post-Soviet Southern Caucasus and Central Asia. Pilot Study: Agricultural Knowledge Systems in Georgia and Tajikistan".This paper was written in frames of this project. Further information: <u>http://www.zef.de/1847.html</u>.

has partially been sustained, partly has been reorganised or dissolved due to the lack of state subsidies (Beniwal et al. 2010). The transformation period has negatively affected the quality of research and education. At present there is no mechanism to link and coordinate knowledge transfer/inflow to agricultural producers from the national agricultural research (done by different organizations) and education (Beniwal et. al 2010; Morgounov and Zuidema 2001). On the district and local levels there are still former *kolkhoz* experts, as in the Soviet period, like agronomists or water engineers who provide agricultural consultations to farmers. In addition, there is a new actor – international donor organisations who provides agricultural knowledge and shapes the knowledge sharing landscape. After the end of the civil war in Tajikistan, numerous international donor organisations came to Tajikistan offering relief development help. International donors as well as NGOs funded by them through different development projects provide all kinds of agricultural inputs and knowledge to farmers. By implementing these projects, many dynamics and interactions emerge between the Tajik government, the international donor community and NGOs. Thus, international donor and development organizations, as well as NGOs (usually supported by donors) and the Tajik government are key players in agricultural knowledge systems (in contrast to the Soviet period), and hence in the rural transformation of Tajikistan.

Central Asia and Tajikistan as part of it have generated global interest due to its energy potential and geostrategic significance (Heathershawa 2011; Jonson 2006; Kreutzmann 2005; Subodh 2003). According to Lena Jonson "Central Asia has become the battleground for the major struggles of the 21st century: radical Islam versus secularism, authoritarianism versus identity politics, Eastern versus Western control of resources, and the American 'War on Terror" (2006). Tajikistan's closeness to Afghanistan has made it an international drug trade route since the early 1990s (De Daniele 2011). Since the late 1990s onwards both the United States and the European Union have been providing assistance to the Tajik government in the field of counter-narcotics (idc). Since then the number of international development projects in different fields has grown in Tajikistan. Amongst other geopolitical projects, development of agricultural advisory services is one of many 'development' projects initiated in the region by the international donor community.

Regarding agricultural advisory services (AAS), many different models exist in the world, each with its pro's and con's. AAS is one of the components of the development projects implemented by international donors in Tajikistan. In order to understand how a national network of agricultural advisory services should be organised in Tajikistan, which is part of the on-going agricultural reform process, I will look at the earlier mentioned knowledge providers, i.e. NGOs and donors, as well as universities, research and former Soviet agricultural experts. However, in order to come up with an appropriate model for agricultural advisory services in Tajikistan, it is not enough to look at existing AAS models and search for 'best practices' that could be copied. What is important is that the model reflects the interplay of the main actors and the local needs. Therefore in this paper I analyse the functioning of and interaction between the international donors, NGOs and Tajik government (and universities, research and former Soviet agricultural experts). After the analysis of the different knowledge providers, I will summarise the main points for constructing an AAS model for Tajikistan.

The paper is organised as follows. In the first two sections I briefly present an overview of the rural economy of Tajikistan after the end of the Soviet Union and the problems farmers face while running agriculture. Section three provides an overview of agricultural advisory services and extension models. Section four outlines the research methodology, followed by Section five introducing the rural economy of Tajikistan. Section six presents the work done in the field of agricultural advisory services by different actors. The same section analyses the interplay of the different actors involved. Based on this analysis, Section seven outlines the features an agricultural advisory service should have in order to meet the needs of different types of farmers and thus to contribute to agricultural development in Tajikistan.

# **3 OVERVIEW OF AGRICULTURAL ADVISORY SERVICE MODELS**

In this paper I study the role of external/global, agricultural knowledge in local knowledge production and further development, as well as the channels through which it reaches local innovators and subsistence farmers. Agricultural extension and agricultural research and education systems were identified as the most important researchable areas in Central Asia and the Caucasus region followed by crop production, water resources, marketing, processing and value addition of agricultural commodities and developing suitable agricultural development policies related to rural employment, land tenure property rights and others (Binewal et al. 2010; GCARD 2010). GCARD underlines the importance of creating knowledge systems and strengthening communication with smallholder farmers and others along the production-consumption chain for participating effectively and equitably in markets (GCARD 2010).

For this purpose I will look at different agricultural advisory service models available. Agricultural advisory service or extension have been organised in different ways in various parts of the world. AAS can be organised based on the University, on farmers' associations, by government agencies on the regional/local levels, or as a combination of these different approaches. In the USA, extension is provided by both, Universities and government. Based on the Smith Lever Act on cooperative extension service (1914), such extension was established linking research, education, land grant universities and county administrative departments (McDowell 2001; Kalna-Dubinyuk and Stanely 2005). The Japanese extension system works mainly through cooperatives (Agbamu 2000; Gereads 2009). Such cooperatives provide marketing, banking, insurance and health care for farmers. Commodity cooperatives are fully funded by the government and play an important role in Japanese agriculture (Agbamu 2000). In the Netherlands, it is a combination of government and professional farmers' associations, i.e. credit, input cooperatives and a mix of the above mentioned with sufficient government support. In Germany as in other industrialised countries, one observes that public extension has been under pressure to introduce cost sharing or altogether commercialize advisory work. An approach which combines commercial and public elements is at present being introduced in some of the eastern states of Germany (Nagel and von der Heiden 2004). For example, the Ministry of Agriculture in Brandenburg subsidizes consultations once they have actually taken place. Thus, the mixed model of extension continues to be one of the popular models under the circumstances of the rapidly changing world (Kalna-Dubinyuk and Stanely 2005).

The Soviet system of knowledge production and sharing differed from the earlier described models (Van Assche et al. 2013 forthcoming). This process was not called extension or AAS, but the knowledge transfer goal was similar. It was organised through a complex network of the Agricultural ministries, agricultural universities, academy of agricultural sciences, research centres, *kolkhozes* and *sovkhozes*. Here knowledge was generated based on the state production targets. Knowledge was generated in the specialised research institutes, which subordinated either to the Academy of Agricultural Sciences or the Ministry of Agriculture (with some variations in different Soviet republics and years) (Morgounov and Zuidema 2001). Later it was channelled to the *kolkhozes/sovkhozes* through the local and regional departments of the Ministry of Agriculture. Every *kolkhoz/sovkhoz* had a set of trained experts, i.e. an engineer, an agronomist, an accountant, etc. who regularly attended classes/courses to update their knowledge. Since the end of the Soviet Union, NGOs fill a big gap of knowledge supply in the agricultural sector of Tajikistan (it will be shown more in detail in the subsequent parts of the paper).

Last but not least, it is important to mention that every model in those countries mentioned was constructed for different farmers. For example, farmers in the USA were autonomous and entrepreneurial in the capitalist system, whereas in the Soviet Union farmers did not exist. *Kolkhoz* management was something that resembles farmers in the US. But with the end of the Soviet Union, *kolkhozes* and *sovkhozes* were dissolved into many different things, same as AAS, as will be presented in the following chapters (Kandiyoti 2003).

# **4 RESEARCH METHODOLOGY**

The data for this paper was collected as part of the Project "Epistemic Cultures and Innovation Diffusion in post-Soviet Southern Caucasus and Central Asia. Pilot Study: Agricultural Knowledge Systems in Georgia and Tajikistan". A literature review and 50 semi-structured expert interviews with different organisations were conducted in various parts of Tajikistan on different administrative levels during two months in 2012. The representatives of the following types of organisations were interviewed: donor and international organisations, Ministry of Agriculture, local and international NGOs, and local state government<sup>2</sup>.

<sup>&</sup>lt;sup>2</sup> The author would like to express her gratitude to all interviewees who spent their time and shared their experience in regard to agricultural advisory services in Tajikistan.

# **5 THE RURAL ECONOMY OF TAJIKISTAN**

There are 7.8 million people in Tajikistan and 70% of them reside in rural areas (Agency of Statistics of Tajikistan 2013; Lerman 2012; Agrodonish 2010). Despite the small share of arable land in Tajikistan, which is estimated at 7% (around one million ha), agriculture is one of the most important sectors of the economy and it provides employment for two thirds of the population. The share of the agricultural sector in the GDP was estimated at 18,7% in 2010 (Tajstat 2011) and agricultural products make up 30% of official exports (FAO 2011). The main agricultural crops are cotton, cereals, grazing for cattle and small ruminants, horticultural crops, potatoes, vegetables, melons, orchards and vineyards (FAO 2011; Boboyorov 2012).

After the end of the Soviet Union, *kolkhozes* and *sovkhozes*, that used to be the main production units, were dissolved. The agricultural reform in Tajikistan is on-going since 1997, attempting to distribute land to farmers. According to the FAO, four types of agricultural producers can be distinguished nowadays in Tajikistan (Lerman 2012). These are (1) household plots, (2) individual and family *dekhan* farms, (3) collective *dekhan* farms and (4) agricultural enterprises (successors of former state farms). Only the first two groups are true family farms. Agricultural enterprises are in fact specialized state-owned farms. Official statistics make it impossible to differentiate between *dekhan* farms of different types. Recent estimates show, however, that collective *dekhan* farms – which constitute less than 5% of all *dekhan* farms in Tajikistan – control more than one-third of the arable land in the *dekhan* farm sector. Despite the significant presence of collective *dekhan* farms, the agricultural sector in Tajikistan is now largely individualized: nearly 65% of arable land is in family farming (household plots, individual *dekhan* farms, and family *dekhan* farms) (Lerman 2012). The farm sector today consists of some 750,000 rural households, each with a household plot of 0.3 hectares on average, and nearly 90,000 *dekhan* farms with 7 hectares of arable land or 30 hectares of agricultural land on average (Ibd).

The end of the Soviet Union and the following civil war had serious consequences for Tajikistan, for agriculture and rural development sectors in particular. Despite reforms in agriculture undertaken by the government, there is a number of obstacles farmers face:

- The mountainous location of most of the country and the poor road and market infrastructure make it difficult to transport agricultural products for sale to the bigger markets of Khujand and Dushanbe. Due to the same reason, there are only little and expensive fertiliser and inputs for agricultural production available (Interview with F. Kurbonov, German Agro Action, April 2012);
- No processing facilities; what used to be there during the Soviet times i.e. wine, tobacco, and juice processing factories is dismantled. Many processing facilities were destroyed during the civil war and little were reconstructed;
- Local governments with their bureaucratic approach to work prevent agribusiness development; As a result of water-related cross-border conflicts with Uzbekistan, the export of Tajik agricultural products to Russia via Uzbekistan is blocked (Juraev 2012; interview with Azamjon, manager of the Khatlon Livelihood support project, April 2012).

Thus farms are small, vulnerable, and under-financed as the following quotation suggests:

'750,000 farms were created, but nothing for this was established. 70% of farm owners are random people, people do not know how to run a farm; no access to fertilizers, no access to markets and marketing facilities, no machinery, very high interest rate for loans in the banks, no insurance in agriculture; high salinity of soils. Nobody carries responsibility for the quality of imported pesticides, herbicides and seeds; low-productivity cattle, no veterinary,

there is no laboratory of quality control. There is nothing' (interview with the director of NGO "Makhbuba", April 2012).

The majority of small scale farmers is poor and they do not have experience in paying for agricultural advisory services. On the other hand, a farmer is not accustomed to pay for consultation, as in the past he used to receive it for free either through the Soviet kolkhoz system or in recent years from donor humanitarian help. Another obstacle for farmers is the lack of agricultural background. Many farmers used to be medical doctors, teachers, accountants and in professions unrelated to agriculture. Similar to the post-Soviet agriculture in Uzbekistan, not all farmers in Tajikistan have learnt how to work, how to make decisions independently without orders from above, how to be profitable, how to manage their own expenses (Shtaltovna 2013). In addition, many men went abroad to look for jobs during the transition period. Thus, left are old people, kids and women. As a consequence, there is a growing number of female farmers nowadays in rural Tajikistan. To be a farmer in Tajikistan today, one needs to become a multi-functional entrepreneur, having specific agricultural knowledge as well as financial and legal skills, marketing skills and being able to find all the needed inputs and machinery (Interview with I. Nematov, director of the NGO 'Bakht', 2012). Given the circumstances presented above, can a farmer reach those skills in some 15 years? Therefore, it is not that easy to become a farmer. A farmer still has to decide, if either he really wants to and can be one. Last but not least, it is not clear at all, if the land reform is completed or if there will be more changes. At the present stage farmers are too weak to boost agricultural production by themselves (Interview with N. Dadabaev, Ministry of Agriculture, May 2012).

# 6 SOURCES OF KNOWLEDGE FOR FARMERS IN TAJIKISTAN

In this section, I will look at available agricultural knowledge and innovation sources for farmers in Tajikistan. This is a mix of remains of the Soviet collectivised agricultural system, Soviet educational system, modern knowledge brought by development agencies and their agricultural projects, and the small scale commercial input suppliers. I will look at the main actors involved in the knowledge sharing sector - international donors, government actors, NGOs, Agricultural University of Tajikistan, and commercial input providers. The activities of these organisations and interactions amongst them are in focus.

As it was presented in the previous section, *farmers* have to overcome many challenges to run agriculture. First of all, many of the present day farmers never were farmers in the Soviet period. There are gaps in knowledge and experience, for example in running agricultural activities, how to be an entrepreneur, how to label a product, prepare a financial report, where to sell farmer's products, etc. Depending on the type of production, either animal husbandry, horticulture, tobacco or cotton production, *kolkhozes* and *sovkhozes* were equipped with a set of trained experts i.e. an engineer, an agronomist, an accountant, etc. Thus *kolkhozes/ sovkhozes* were independent and fully equipped production units. In addition, there used to be a state agricultural unit on the district and regional level called 'Agroprom' (a regional department of the Ministry of Agriculture). This organisation was responsible for the functioning of *kolkhozes/ sovkhozes*, thus it also provided agricultural consultations.

With the end of the Soviet Union, there are no more *kolkhozes*, instead there are thousands of farmers with land certificates, many of whom have not actually received land yet (see Section 5). Farmers' main source of knowledge are agronomists from *Agroprom* or former *kolkhoz* specialists, however their expertise is outdated and does not always meet the present day farmers' needs. Farmers do not expect help from the state, as there is hardly any available. For instance, the state agronomist has no opportunity to visit farmers' fields or call them because he receives no funds for this purpose. Therefore, farmers have to visit him in the district capital in case they need a consultation (Interview with the First Deputy Mayor in charge with agriculture of the Penjikent municipality office, April 2012). In addition to the underpaid state agronomist, the former *kolkhoz* agronomists and a few other former *kolkhoz* workers can provide knowledge to farmers in the rural areas. Very often these former *kolkhoz* agronomists are employed by international organisations or NGOs to work for agricultural projects implemented by donors in rural areas of Tajikistan.

Cooperation between NGOs, donors and the local government (*jamuat*) occurs on the issues of agricultural extension. Many of the government organisations remain thinking and working in the old Soviet style, meaning a top-down working approach, giving commands, controlling and reporting.

"Management in agriculture is destroyed. The government tries to solve new problems with the Soviet approaches and experience" (Interview with A. Sharipov, head of the National farmers' association, May 2012).

From years of cooperation with local governments, NGOs in the southern part of Tajikistan have had all kinds of experiences. The below citation represents the way how they usually work

"Due to the low state budget, jamuat [local state administration] representatives go around the village and collect money from the villagers to celebrate Navruz or May 9<sup>3</sup>. They collect it amongst the farmers. But where can the farmer get money? I tell them give me a receipt and I will pay you. After that, they do not come back to me again. Those who are afraid and

<sup>&</sup>lt;sup>3</sup> National holidays in Tajikistan

don't know, they keep on paying. I am fed up, I don't want to deal with government organisations any longer!' (Interview with A. Mamadshoev, agricultural expert at Mercy Corps, May 2012)

There are also cases when the activities of the NGOs financed by international donors are perceived as a source of financing for the local government and a power competitor in rural areas. As indicated by the following:

"There is just this (one) agro shop in the area. We asked the state administration to give us a room for a shop, it was refused. Later, that premise was sold. Moreover, the state officials always ask for a tractor and do not pay for it, threaten and do not support our activities at all" (Interview with the director of the NGOs' Union "Iftikhor", May 2012).

However, there is a difference in how district/regional government works in the south and in the north of the country. In Khujand region I have seen more positive or neutral experiences of cooperation between the state and NGOs. Here NGOs reach higher goals in contrast to the south of the country in terms of cooperation with the state. Experienced NGOs like "SAS Consulting", "NAU Khujand", "Zarzamin" manage to successfully cooperate with the state on the issues of seeds import to Tajikistan and reduction of added value tax for the import of fertilisers and seeds. What one can expect from the cooperation with the local, district or regional government is non- disturbance of both sides, also the government is good at mobilizing the public and organizing a room if an event is to be conducted in some other area. Basically, government has not much to offer, it is the other way around, and they sometimes appeal for help to NGOs (Interview with the director of the agricultural advisory service NGO 'Zarzamin', May 2012).

In the past 10-15 years farmers received inputs and knowledge through international projects targeted at food security and agricultural development. International donors are relatively new actors in a traditional society like Tajikistan. They came to Tajikistan to provide relief help after the civil war in 1997. From 1998 till 2002, development agencies focused on providing and distributing inputs to farmers free of charge. Starting from the early 2000s, they have shifted from distributing inputs for free to distributing inputs for partial payment (interview with M. Nurmatov, extension expert at USAID, May 2012). And as part of their work in the region, these organisations provide inputs and knowledge to agricultural Agricultural advisory systems or extension of the conventional type to support farmers producers. did not exist during Soviet times. New forms of extension systems to support small holder farmers and producers, very different from the collective farms of the Soviet period, are still not developed; they take their time to emerge. In the past years development of agricultural advisory services often was an element of agricultural and rural development projects financed by international organisations. Having in mind the serious problems in rural areas and the importance of agriculture, the provision of agricultural advisory services is a significant contribution of the development projects and NGOs in the rural development sector of post-Soviet Tajikistan.

Presently there are approximately 200 different international donor organisations working in Tajikistan. They implement a big variety of agriculture-related projects, providing different services in different parts of Tajikistan with duration of 2-5 years. The international organisations where agricultural advisory service is one of the leading agricultural projects are the European Bank for Reconstruction and Development, USAID, The World Bank, Aga Khan Foundation, FAO, JIKA, British embassy, OSCE, Mercy Corps, Swiss Embassy for development and cooperation, GIZ, Oxfam, and others.

However the aid is uncoordinated and just few international organisations have a long-term strategy (International Crisis Group 2003). Usually donors neither cooperate amongst each other and with other present NGOs, nor coordinate their activities in case they work on a similar issue. This sometimes results

in an overlap of donors' and NGO's activities. There is also no clear picture of what has been implemented, by whom, when and what is left after the projects.

"A Swedish organization has set up 3 laboratories on seeds breeding, but we don't know where they are located. Neither state or farmer, nor the consultants know where it is" (Interview with Muhiddin Sharipov, Ministry of Agriculture, May 2012).

By means of development projects, international donors provide all kinds of knowledge, inputs and services to farmers. Sometimes the expertise and inputs provided by donor organisations do not meet the needs and reality of the Tajik farmers, as the following quotation suggests:

"If it is donors, they work for food security and tell farmers to grow other kind of crops. We don't know how to grow them, there is no market where to sell them or where to process them.....They say it is a commercial crop and you should grow crops for your own production. And people don't know how to preserve, to process and to store it" (Interview with A. Mamadshoev, agricultural expert at Mercy Corps, May 2012).

Thus there are situations when a farmer does not have knowledge on how to grow the new crops suggested by the development agency. Neither is there a market where farmers can sell the agricultural products, nor processing facilities.

Too much help from donors towards farmers has rendered farmers to be not eager to pay for consultations or find solutions for their problems on their own:

'For the past 15 years farmers are fed by donors' help. 90% of rural population live thanks to the donors help ('90% сельского населения сидят за счет донорской помощи'). And it is hard for a farmer to move from it. The first question the farmer gives, when he sees any organization is 'what will you give us?' (Interview with the representative of NGO 'Zar-zamin', May 2012).

It is also important to mention, that the knowledge and innovation component is not the main objective of many projects, it is rather a secondary goal. Some donors themselves pursue geo-political goals in the region after the end of the Soviet Union and agricultural development projects are just a small share of their activities in the region. Also, the origin of each donor often determines the approach chosen for project implementation. German, American and Japanese development cooperation have different approaches, ideologies and traditions, as indicated below:

"For instance, for Americans the situation in Tajikistan is democratic in contrast to what is going on in Iraq and Uzbekistan" (interview with G. Tolibzoda, director of the extension service 'MMK/ATAK', May 2012).

JIKA, the Japanese development cooperation, has implemented their projects through the national farmers' association because the similar association in Japan is very strong. Thus, it is trying to reproduce successful examples from Japan in Tajikistan (Interview with S. Karimov, former JIKA consultant, April 2012).

*Many NGOs have occurred to implement donors' projects*. At present, there are approximately 1500-2000 NGOs in Tajikistan (Agrodonish 2012). The Emergence of NGOs is a response to problems in the society caused by the civil war and dissolution of the Soviet Union, with no state institutions which would solve these problems. Most of them were established to fulfil donors' projects. International projects and donors play a big incentive for, and mobilise a lot of actors through, the established NGOs. Some NGOs became a platform for active and knowledgeable people, former socialist specialists in diverse fields of knowledge and expertise.

NGOs are filling a big gap in the rural economy of Tajikistan. They provide not just advisory services, but there are many initiatives implemented through the projects financed by donors. For example there are projects on improving the law on farms (financed by USAID), working with farmers and developing the land code (with OSCE) and a law on cooperatives and other issues in agriculture; community mobilization (Mercy Corps), preventive work and work with women; farmers, on the other hand, have a very low level of awareness about their rights in regard to the land reform. One of the important problems that agricultural advisory services help to solve is bridging farmers with processors and local markets. For this deal agricultural advisory service receives a small percentage of the benefits. By implementing development projects, NGOs and their actors are also trying to solve many other problems in the socio-economic transformation process rather than just providing agricultural advice. NGO's with their projects try to bring government representatives to solve farmers' problems (in relation to the land reform). Donors (with their money) play a big incentive and mobilise many active and knowledgeable people who unite in and act via NGOs in the rural areas. Also donors provide a strong backup for the NGO to start acting. Without donors they would not be so courageous and proactive.

# **Table 1:** Selected examples of NGOs working on agricultural advisory services that are mainly financed by international donors

Organisation	Method & approach to knowledge transfer	Targeted
organisation		agricultural
		produce
Agricultural	The AIN network consists of 5 district centres in Sogd region (north) and 6	bee-keeping,
information	centres in Khatlon region; each has appr. 5 distirct centres, so around 30	drying fruits and
network (AIN)	small local consulting centers in Tajikistan. In each local centre there is a	vegetables, how
network (Ain)	vet and an agronomist. In addition, every centre has one of the following	-
	innovative consulting products and packages: on bee-keeping, drying	to grow rabbits, ostriches, sun-
	fruits and vegetables, how to grow rabbits, ostriches, sun-choke. And	choke
	whenever his/her package/expertise is needed in one of the centres, they	CHOKE
	exchange their expertise, send their consultant to another part of Tajikistan	
	Tajikistan	
SAS Consulting	An sms consultation service on different crops: from choosing a crop to	All possible crops
	harvesting.	
	Farmers are ready to pay for demonstrations on	
	<ul> <li>How to grow tomatoes,</li> </ul>	
	<ul><li>melons,</li><li>to protect from apricot pests,</li></ul>	
	<ul> <li>how to construct and take care of a greenhouse,</li> </ul>	
	<ul> <li>nursery/ seedlings</li> </ul>	
	Farmers are already paying for	
	<ul> <li>Tillage</li> </ul>	
	<ul> <li>Sowing</li> </ul>	
	<ul> <li>Harvesting and transportation of the agri products;</li> </ul>	
	<ul> <li>Sowing grains</li> <li>It places bridging role between formers and processors and for that</li> </ul>	Wheat grapps
ΜΜΚ/ΑΤΑϹ	It plays a bridging role between farmers and processors and for that	Wheat, grapes,
	ATAC receives %. qualitative consultation, theoretical and practical	orchards, potato,
	trainings, introduction of innovations and other services in the field of	sunflower,
	agriculture	onion, pee,
	<ul> <li>Integrated Production Management (IPM)</li> <li>Farmers Field Schools (FFS)</li> </ul>	fodder, cattle,
	<ul> <li>Village Advisor system (VA-system)</li> </ul>	etc.
Agrodonish	Agrodonish is a national forum of advisory services. 6 Agricultural	
0	advisory networks are members of Agrodonish.	
	www.aist.tj	
NGO 'Shifo'	Project financed by the World Bank: Dealing with the cotton debt. It	chicken, lucerne,
_	encompasses trainings on 'the rights of <i>dekhan</i> farmers' and 'efficient	tomato seeds,
	management of the <i>dekhan</i> farm'.	potatoes, wheat.
	Another project 'Food security: seeds import'. We distributed seeds in	,,
	two districts – Djaldy Rushi (58 farmers) and Shahrituz (50 farmers).	
	October 2010 we started distributing eggs and forage to 108 groups. 15	
	chicken groups – 15 persons per group. When chicken are born, they are given to another 3 persons in the group and so on. We also distribute	
	lucerne for 5 ha for 6 groups, in total for 30 ha.	

The essential factor for every NGO to survive is financing. Donor organisations are so far the only source of money for NGOs. For any NGO or advisory service it is hard to survive alone, as farmers are not ready to pay for consultations yet (Interview with Dr. N. Mirzoev, director of the Agricultural Information Network, May 2012). Moreover, there is no state support for extension, thus hardly any money is allocated in that direction. Many NGOs do not see how they can survive as organisations when there is no donor money. As the following citation from the interviews suggests

'The donors' assistance is needed. Through donors our salaries are covered for 95% and 5% are covered through farmers' payments' (Interview with M. Otobekov, agronomist at SAS Consulting, May 2012).

In order to survive and to be self-sufficient, organisations that provide advisory services adopt different strategies. Thus, they provide a number of other functions and services like machinery, milling, linking to the market, linking to the agricultural inputs, with salesmen and processors, etc. In this way, it is requested to fill the gap left by all services which previously were provided by the government and through the system of *kolkhozes* and *sovkhozes*. For example, the agricultural advisory service 'MMK/ATAC' has, in order to sustain when the donors' funding is not there, started a guesthouse. Another organisation, 'SAS Consulting', that was earlier supported by the World Bank, also had to start up additional activities in order to sustain as an organisation. So "SAS Consulting" for example has established its own micro-financing organisations. Thus, some organisations have appealed to diversify their activities in order to maintain functioning as an organisation rather than giving up and thus losing the gained experience and established contacts with farmers, local communities, district and regional government.

There are also cases when NGOs change their activities, following the donors' calls, as indicated by the head of NGO 'Agrodonish':

'Today an NGO is involved in medical services, tomorrow in extension. Can one consider it to be an extension service and invite it to be a member of an advisory service network?' (Interview with U. Kasimov, director of NGO 'Agrodonish', April 2012).

Thus, all these organisations that emerged as implementers of donor-funded agricultural projects create strong competition (while receiving money from donors and providing service to farmers). Many donors continue to give funds just through their established NGO network, excluding sometimes more competitive organisations or consultants.

When talking about the establishment of a nation-wide agricultural advisory services network, some NGOs are cautious about participating in it. The reason is that smaller NGOs are afraid to share their ideas, as there are examples from the past when the ideas were collected and the stronger ones started ruling without involving the smaller ones or bringing them anything in return; moreover the Soviet ruling background is very fresh. They mean:

Maybe in 10 years we can try, but not now, not yet (Interview with S. Ghulomhaydarov, agricultural expert at NGO "Shifo", May 2012).

Thus, many smaller NGOs do not trust in cooperation with the present transition state.

Some NGOs become trapped and abused by other, bigger and more powerful NGOs or state organisations. A boss of a local rural NGO tells the following

"XYZ organizes people, a room and sometimes a small coffee break for the NGO "X", who is financed through CVB. Some people call her sometimes at night and ask to organize people and a room for tomorrow. She has been doing it for 3 months. She is not paid. She wouldn't do it, but she can't say 'no'. Sometimes she tries not to pick up the phone, but every time

they call from the new number. What they teach is public health, HIV/AIDS prevention, etc. people have attended already this kind of trainings and many of the participants express dissatisfaction. They do not organize any coffee breaks; do not provide any materials, paper or pens. They always bring a box of pencils and collect them after the seminar. They always bring this one box of pencils. Once XYZ could not help them, but they asked her 'oh, give us someone who could help us'. So she asked her friend to help. And later, this friend of hers, has reported that they asked her to put in the list extra fake names of participants, so instead of 15 participants, 30 were reported' (interview with XYZ, May 2012).

These events unfortunately take place in order to obtain donors' funding.

There are also successful examples of **cooperation amongst different NGOs and donors.** Donors and NGOs try to establish a space for exchange of innovations and achievements, for example, an **Agroplatform and study laboratory**. It was established by Helvetas for all interested parties to share their attainments and learn from each other. Each time, a topic is chosen and one organization presents its results. These are SAS - Consulting, CESVI, Zarzamin, Saodad, NAU. But this takes place on a small scale and does not involve all actors dealing with knowledge and innovation.

There is an example of the win-win relationship between the donors and NGOs. For example, between the Agricultural Information Network (AIN), the international NGO 'ACTED' and the national association of agricultural advisory services of Tajikistan "Agrodonish". ACTED is the only organization that supported the NGO 'Agrodonish' and 'AIN' (Interview with R. Mansurova, director of ACTED, May 2012). ACTED provided both NGOs an office space free of charge and advertising of their activities. On the other hand, AIN and Agrodonish usually play the role of partner organisations for ACTED in terms of their activities and application for funding.

Another potential knowledge source would be the *Academia* including agricultural universities, colleges and research centres where experts and special agricultural knowledge were produced and later disseminated to the production units (Morgounov and Zuidema 2001). Tajikistan inherited from the former Soviet Union high quality agricultural research and education systems with strong linkages between one another and a wealth of research experience from the Soviet era (Beniwal et. al 2010). Furthermore, there used to be an 'Association of Knowledge' (<u>http://www.znanie.org/OZR/history.htm</u>) that provided advice for agriculture upon request. Despite this provided fairly good foundation to build-on, collaborative agricultural research for development after independence and the agricultural research and education system has suffered due to insufficient support provided by the government and the exodus of the qualified people after 1991 (Beniwal et. al 2010, Morgounov and Zuidema 2001). As a result, many former scientists are ageing and their expertise is becoming outdated, or has left the country. And the young researchers are not motivated to work in academia because of the low salaries. The linkages through which agricultural research could contribute to innovation, the well-being of producers and the economy of the state have been lost.

Young people do not want to study at the Agricultural University (Interview with A. Ahmatov, agricultural expert at Welt Hunger Hilfe, May 2012). The main reasons for that are that knowledge is outdated and that it is hard to find a well-paid job afterwards. Students after graduation do not return to the countryside to run agriculture. Moreover, young men go to the Agricultural University not to obtain agricultural education, but to avoid army service (Interviews with A. Ahmatov, agricultural expert at Welt Hunger Hilfe, May 2012).

In order to provide agricultural advisory services NGOs, who are the main actors nowadays, appeal to the agricultural universities and research institutes. Many Universities cooperate with NGOs in frames of international projects. For them it is a way to share their knowledge as well as to sustain through the transition. The research institute offers its research facilities as well as expertise and the NGO bridges it

with farmers. In addition, NGOs show research institutes (Michurin institute in Sogd region) how to work with farmers, as before they didn't have to do this (interview with M. Safarova, director of the NGO 'EHIO Farhang va Tarakkiyot", May 2012). What farmers want to see is practice, but the University has limited resources and possibilities. Universities are not that proactive and do not contact NGOs to cooperate first. Partially because they do neither have such experience, nor money to initiate new research projects, nor new high-tech to offer.

As to the *commercial input and seeds suppliers*, there are not many, but they are available on the openair markets in the district and regional centres. However, it was often mentioned, that input sellers on the local markets are usually not experts of their business, thus they hardly can guarantee the quality of the inputs and seeds they are selling (interview with A. Ahmatov, agricultural expert at Welt Hunger Hilfe, May 2012).

To summarise this part, there are different actors present and many things are happening in terms of knowledge production and sharing in agriculture. By observing agricultural development projects provided by the donor community, we see how these projects became a battle field for obtaining all kinds of interests, especially from the government and the newly emerged "NGOS". Donors are new actors in the knowledge and development field, but important ones. Under the framework of 'development', i.e. the rhetoric, organization and infrastructure of development, different donors play their own games, some of them geo-political, as it was illustrated in section two. At the same time donors provide a political backup for action and provide an opportunity to sustain for local NGOs. Last but not least, donors are used by Tajik political actors also to sustain organisationally and individually in the period of transition. As to the farmers, due to a lack of agricultural background and no big choice of knowledge sources and little governmental programs to support agriculture, they are very often ready to take what is there. Thus they are not critical about which kind of knowledge they receive from various sources.

# 7 DISCUSSION: HOW SHOULD AGRICULTURAL ADVISORY SER-VICES LOOK LIKE IN TAJIKISTAN?

In this paper I looked at knowledge systems and channels of innovation in Tajikistan. Due to the end of the Soviet Union and the devastating civil war in the 1990s, agriculture, which is one of the main employers for the rural population, undergoes changes and difficulties. At the present day 750.000 farmers do not have farming experience and knowledge on how to run agriculture; neither is there an infrastructure for running agriculture including markets, roads, and input supply. Presently knowledge sources to farmers can be described as a mix of former Soviet traces and fragments originating from the Soviet agricultural and educational system represented by the universities, research institutes and the academy of sciences; and on the other hand, western-style knowledge, mainly introduced by development agencies. Thus, there are government, NGOs and international donors who are the major knowledge sources for the Tajik farmers.

Agricultural advisory services are one of the promising knowledge sources for farmers that occurred as a result of transition. However while looking back at the presented models of AAS in different countries, hardly any of them can fit Tajikistan. The government was an important actor in establishing AAS in the USA and the Netherlands. After the end of the civil war the situation in Tajikistan was destabilised and with the present economic problems it is even more undermined. It tells us something about the position of the government. In regard to the AAS, the Tajik government tries to develop AAS based on the Soviet principles and available structure. However, not many actors, especially NGOs and donors, agree with that approach, and there is no way back to Soviet times. The Tajik government showed little interest, i.e. allocation of resources to establish AAS. What we saw, was more the individual interests of the government representatives to maintain their positions at the expense of donors' money. Due to its

poor situation, the government tries to stay alive by sometimes inventing new functions. Sustaining with government resources is sometimes more important for the governors than their primary goals.

The active actors of AAS in Tajikistan proved to be NGOs and donors. As we could see, many NGOs have emerged and somehow fill this gap of knowledge and many other public services which have dissolved after the end of the Soviet Union. But NGOs are not the government and cannot decide alone on how to organise AAS, also because they will not stay for ever. The government very often does not approve what is suggested by NGOs. Due to a big number of NGOs there is a coordination problem. Also due to the fact that financing for NGOs mainly comes from international donors, there is a competition for money amongst NGOs. Thus, there are conflicting ideologies and interests of the government, donors and NGOs who once talked about establishing a nation-wide agricultural advisory service. In addition there are donors with their own traditions, ideologies and agendas. For example, JIKA was trying to establish AAS through the farmers' association, as it works in Japan. Also a farmers' association played a crucial role in establishing AAS in the Netherlands or Belgium. But in Tajikistan, this association is very weak and has no political strength. Moreover, it was



An illustration from the Krylov's fable "Swan, Pike And Crawfish".

Image source: http://www.slideshare.net/VictorZinger/i van-andreyevich-krylov one of the examples when the NGO was hunting for donors' money. As it was mentioned, the dense presence of the international donor community who mainly support AAS in Tajikistan can be connected to the closeness of the Afghan border. If in 2014/2015 the army will leave Afghanistan, it can very likely happen that many donor organisations will stop their missions in Tajikistan as well. This will have a direct impact upon establishing AAS in Tajikistan. And this makes the situation for establishing AAS very unstable. Thus this approach does not seem to be applicable in Tajikistan. None of the suggested models of agricultural advisory services can fit Tajikistan. Moreover, AAS is a new structure in post-Soviet Tajikistan and its existence remains under a question mark.

To this end, AAS system development depends more on the actors involved in the decision making process, rather than choosing one or another extension model (Kalna-Dubinyuk and Stanely 2005). The actors currently involved in AAS in Tajikistan so far are reminiscent of Krylov's fable, where the individual interests and agendas of the actors involved are more important than a mutual goal - establishing an AAS in Tajikistan. Thus, a national-wide AAS is being discussed since the late 2000s and has not been established since then.

Against this background, I suggest the following to take into consideration once designing national AAS fitting Tajik realities. First of all, AAS should build on existing assets, traditions, networks and on the available capacities, i.e. human capital, expertise, and physical infrastructure, knowledge sources and experience of different NGOs and donors. Secondly, a favourable environment to run agricultural business is a necessary condition to first develop and then run agriculture. A broader picture should be taken into account that is legislation encouraging taxation of agricultural producers, laws on private entrepreneurship, access to land, corruption, security, access to credit, having certificates on land, machinery, and market infrastructure. If there is a market where farmers can sell their products, then farmers would be interested to produce and just then they might need AAS (Interview with W. Van Weperen, extension expert at Caritas, April 2012). Thirdly, the background and expertise of agricultural consultants are crucial. Farmers want to work with professionals and people holding the right qualifications. So far, all people who are involved in agricultural service provision through NGOs and international organisations, have a diverse background, varying from former kolkhoz agronomists and mechanical engineers to medical doctors and economists. Also, taking into consideration the poor situation at the Agricultural University and agricultural colleges, the agricultural experts need retraining, capacity building, they should be taught how to mobilise local administration for successful cooperation and others. To this end, an agricultural consultant should be motivated in what he is doing, stay in touch with farmers. The qualified and motivated staff of AAS will be an incentive for farmers to pay for agricultural consultations. Therefore, a permanent agronomist/agricultural consultant on the local level would be crucial for agricultural and rural development.

Fourth, cooperation between NGOs, donors and the state is important for AAS development. The government keeps a grip on everything that is taking place in Tajikistan, especially when there are international funds involved. Therefore AAS can build on existing state administrative structures stretching from national, through regional, district to mahalla (local) level and available experts (i.e. agronomists) who are already there. However, without applying old Soviet working methods, which proved to be outdated and undesired by other actors involved (i.e. NGOs, donors, and farmers). There are examples of cooperation between NGOs, donors and local government. They show how the local government becomes more active through this kind of cooperation. The local state administration should be encouraged to cooperate and contribute what it does well, i.e. organising and mobilising local people. Moreover, while working on the grass-root level, one gets a better picture of the local situation, conditions and actors. And here the local government can provide a useful support.

*Fifth*, AAS should be designed based on the needs of **different categories of farmers** presently existing in Tajikistan (see Section 3). **Big and strong** agricultural producers have enough resources to afford a pri-

vate consultant/agronomist or carry out market research. Wealthy farmers are involved in the cotton sector and animal farming. These two categories of farmers almost do not need any assistance from AAS, although they are probably the main category of farmers who are able to pay for advisory services. The *dekhan* farmers are probably the biggest group of the potential AAS clients. On the one hand, they need innovations for production and on the other hand, they have to be interested in paying for consultations. For that, AAS should properly study farmers' needs in order to offer them the right product. AAS should mediate and assist farmers in linking them to the markets, input providers, processers, marketing and other services. It should be demand-oriented and decentralised. Talking about the family farmers, the smallest and most vulnerable category, they need advice but they are not ready to pay for it. It is hard to provide them with any consultation as there are many and they are spread over the hardly accessible areas in Tajikistan. What they grow is mostly for self-consumption, they can hardly buy new seeds or fertilisers and they are not ready to pay for agricultural advisory services. There is an increasing demand from female farmers (dekhan and family farmers) to receive advisory services provided by female AAS staff. In this way they could be more open and have a better interaction with the consultant. Therefore, there is a need for a system where the dekhan farmers and family farmers can benefit and access AAS.

Sixth, a national AAS should provide other services apart from agricultural advice as well as combine different approaches in order to meet the needs of different farmers as well as to sustain without external financing. It should offer both, innovations and advice on the traditional crops. Both should fit well the given geo-climatic conditions of the locate (while suggesting crops to grow, the local natural conditions, available market for the product as well as available processing capacities have to be thoroughly considered). There is a vast experience from the past as well as from the international community in Tajikistan to learn from. In Soviet times, almost all services needed for the functioning of a kolkhoz were part of it, apart from bigger machinery and fertilisers (Shtaltovna 2013). There are successful cases of AAS models working in Tajikistan provided by GIZ (TAG), Helvetas (product chain), Oxfam Novib (agroinput shops), SAS Consulting (agricultural consultation via sms-service), and also examples of business incubators from the Ukrainian and Moldavian model (where farmers are members of AAS) (DAI 2011, Agrodonish 2010). Provision of the AAS can be combined with the processing company, with the consultation at the agro input shop, together with renting machinery services, bio-labs, info-consulting centres, and together with Water Users' Associations. AAS should link farmers with processors, purchasers, with agricultural businessmen and wholesalers while the service should be included into the price of the above mentioned services. While combing agricultural advisory with other services, AAS offers the missing services in Tajik countryside and it manages to sustain its organisation on the longer run without donor support. If the farmer gets a better harvest after he received a consultation, he is willing to pay for the advisory service next year (Interview with M. Suleymanova, director of SAS Consulting, May 2012).

# **8 REFERENCES**

Agbamu, J. U. (2000): Agricultural Research–Extension Linkage Systems: An International Perspective. Agren Network Paper No.106.

Agency of Statistics under President of the Republic of Tajikistan (2013) http://www.stat.tj/en/

Agrodonish (2010) Agricultural extension in Tajikistan, Oxfam Novib, Dushanbe.

- Beniwal, S., Maru, A., Khalikulov, Z. and Ahmadov, H. (2010) Central Asia and the South Caucasus: Challenges, Opportunities, Priority Needs and Actions Required for Improving Agricultural Research for Development. Central Asia and the Caucasus Association of Agricultural Research Institutions (CA-CAARI). pp. 1-57
- Bliss, F. (2012) 'Agrarian Reform of the Republic of Tajikistan A Roadmap for Local Governance and Agriculture Management Institutions Reform' FAOUN in Dushanbe, Tajikistan.
- Boboyorov, H. (2013) Collective Identities and Patronage Networks in Southern Tajikistan, ZEF Development Studies Series, Münster: LIT Publishing House.
- Boboyorov, H. (2012) "Masters and Networks of Knowledge Production and Transfer in the Cotton Sector of Southern Tajikistan". Working Paper Series 97, Bonn University, Center for Development Research (ZEF), pp. 1-24.
- DAI/Petra Geraedts (2011) Survey of existing agricultural extension providers and related programs in Tajikistan. USAID Family Farming Program, Tajikistan.
- De Danieli, F. (2011) Counter-Narcotics Policies in Tajikistan and Their Impact on State Building. In: Central Asian Survey (2011), Vol. 30, No. 1. Special Issue: Tajikistan: The Source of Statehood
- FAO and Ministry of Agriculture (2011) ,Crop and Food Security Assessment'. Mission Report of World Food Programme. <u>http://www.fao.org/countryprofiles/index/en/?iso3=TJK&subject=3</u>
- GCARD (2010) Transforming Agricultural Research for Development. Synthesis Report, pp. 1-20
- Gereads, P. (2009) Report on Workshop "International Workshop on Extension in the Agricultural Sector in Tajikistan", Dushanbe, Tajikistan, pp. 110.
- Heathershawa, J. (2011) Tajikistan Amidst Globalization: State Failure or State Transformation? In: Central Asian Survey. Special Issue: Tajikistan: The Source of Statehood. Vol. 30, No. 1
- International Crisis Group (2003) Tajikistan: A Roadmap for Development, Executive Summary and Recommendations. Asia Report N°51, Osh/Brussels.
- Jonson, L. (2006). Tajikistan in the New Central Asia: Geopolitics, Great Power Rivalry and Radical Islam. I.B.Tauris London. pp. 1-252
- Juraev, S. (2012) Central Asia's Cold War? Water and Politics in Uzbek-Tajik Relations. PONARS Eurasia Policy Memo No. 217. American University of Central Asia (Bishkek), pp. 1-5
- Kalna-Dubinyuk, T., Stanley, J. R. (2005) The Development of Extension Service in Ukraine and the World Experience. Agrarna Nauka, Kyiv, pp. 1-200.
- Kandiyoti, D. (2003). "The Cry of Land: Agrarian Reform, Gender and Land Rights in Uzbekisan " Journal of Agrarian Change Vol. 3(Nos. 1 and 2): pp. 225-256.
- Kazbekov, J., Qureshi, A. S. (2011) Working Paper 145 'Agricultural Extension in Central Asia: Existing Strategies and Future Needs' by IWMI, Vol. 145, pp. 1-35.
- Kreutzmann, H. (2005) The Significance of Geopolitical Issues For Development of Mountainous Areas of Central Asia. Paper presented at International Workshop "Strategies for Development and Food Security in Mountainous Areas of Central Asia", Dushanbe, Tajikistan, June 6-10, 2005.

- Lerman, Z. (2012), Agrarian Reform of the Republic of Tajikistan. Farm Reform and Restructuring Cooperative Development'. Report. Dushanbe, Tajikistan, pp. 1-38.
- Mandler, A. (2013) "Knowledge and Governance Arrangements in Agricultural Production: Negotiating Access to Arable Land in Zarafshan Valley, Tajikistan". Working paper Series 106, Bonn, University of Bonn, Center for Development Research (ZEF), pp. 37.
- McDowell, G. R. (2001) Land Grant Universities and Extension into 21<sup>st</sup> Century: Renegotiating or Abandoning a Social Contract. Ames, Aiowa State University Press.
- Morgounov, A., Zuidema, L. (2001) The Legacy of the Soviet Agricultural Research System for the Republics of Central Asia and the Caucasus
- Nagel, U. J., von der Heiden, K. (2004): Germany- Privatizing Extension in Post-Socialist Agriculture—The Case of Brandenburg. In: William Rivera/ Gary Alex (editors): Extension Reform for Rural Development- Volume 2. Privatization of Extension Systems. Case Studies of International Initiatives. The World Bank: Agriculture and Rural Development Discussion Paper 9. S. 30-35.
- Nikonov, A. (1995) The Spiral of the Centuries-Old Drama: Agricultural Science and Policy in Russia (XVIII XX centuries). Publisher: Entsiklopediia rossiiskikh dereven.Text in Russian
- Shtaltovna, A., Mandler, A. (2012) Workshop Report on 'Agricultural Knowledge Governance and Advisory Services in Post-Soviet Tajikistan' Date: April 17-18, 2012, Venue: Agrarian University of Tajikistan, Dushanbe, Tajikistan.
- Shtaltovna, A. (2013) 'Servicing Transformation: Agricultural Service Organisations and Agrarian Change in Post-Soviet Uzbekistan'. ZEF Development Studies, LIT Verlag, pp. 1-214.
- Subodh, A. (2003) Central Asia Geopolitics and U.S. Policy in the Region: The Post-11 September Era. In: Mediterranean Quarterly (2003): Vol. 14, No. 2. pp. 95-109.
- Van Assche, K., Hornidge, A.-K., Shtaltovna, A., Boboyorov, H. (2013 forthcoming) Epistemic Cultures, Knowledge Cultures and the Transition of Agricultural Expertise Rural Development in Tajikistan, Uzbekistan and Georgia. In: Working Paper Series XX, Bonn University, Center for Development Research (ZEF), XX pp.

ZEF Working Paper Series, ISSN 1864-6638

Department of Political and Cultural Change

Center for Development Research, University of Bonn

Editors: Joachim von Braun, Manfred Denich, Solvay Gerke, Anna-Katharina Hornidge and Conrad Schetter

- 1. Evers, Hans-Dieter and Solvay Gerke (2005). Closing the Digital Divide: Southeast Asia's Path Towards a Knowledge Society.
- 2. Bhuiyan, Shajahan and Hans-Dieter Evers (2005). Social Capital and Sustainable Development: Theories and Concepts.
- **3.** Schetter, Conrad (2005). Ethnicity and the Political Reconstruction of Afghanistan.
- 4. Kassahun, Samson (2005). Social Capital and Community Efficacy. In Poor Localities of Addis Ababa Ethiopia.
- 5. Fuest, Veronika (2005). Policies, Practices and Outcomes of Demand-oriented Community Water Supply in Ghana: The National Community Water and Sanitation Programme 1994 2004.
- **6.** Menkhoff, Thomas and Hans-Dieter Evers (2005). Strategic Groups in a Knowledge Society: Knowledge Elites as Drivers of Biotechnology Development in Singapore.
- 7. Mollinga, Peter P. (2005). The Water Resources Policy Process in India: Centralisation, Polarisation and New Demands on Governance.
- 8. Evers, Hans-Dieter (2005). Wissen ist Macht: Experten als Strategische Gruppe.
- **8.a** Evers, Hans-Dieter and Solvay Gerke (2005). Knowledge is Power: Experts as Strategic Group.
- **9.** Fuest, Veronika (2005). Partnerschaft, Patronage oder Paternalismus? Eine empirische Analyse der Praxis universitärer Forschungskooperation mit Entwicklungsländern.
- 10. Laube, Wolfram (2005). Promise and Perils of Water Reform: Perspectives from Northern Ghana.
- **11.** Mollinga, Peter P. (2004). Sleeping with the Enemy: Dichotomies and Polarisation in Indian Policy Debates on the Environmental and Social Effects of Irrigation.
- **12.** Wall, Caleb (2006). Knowledge for Development: Local and External Knowledge in Development Research.
- **13.** Laube, Wolfram and Eva Youkhana (2006). Cultural, Socio-Economic and Political Con-straints for Virtual Water Trade: Perspectives from the Volta Basin, West Africa.
- 14. Hornidge, Anna-Katharina (2006). Singapore: The Knowledge-Hub in the Straits of Malacca.
- **15.** Evers, Hans-Dieter and Caleb Wall (2006). Knowledge Loss: Managing Local Knowledge in Rural Uzbekistan.
- **16.** Youkhana, Eva; Lautze, J. and B. Barry (2006). Changing Interfaces in Volta Basin Water Management: Customary, National and Transboundary.
- 17. Evers, Hans-Dieter and Solvay Gerke (2006). The Strategic Importance of the Straits of Malacca for World Trade and Regional Development.
- **18.** Hornidge, Anna-Katharina (2006). Defining Knowledge in Germany and Singapore: Do the Country-Specific Definitions of Knowledge Converge?
- **19.** Mollinga, Peter M. (2007). Water Policy Water Politics: Social Engineering and Strategic Action in Water Sector Reform.
- 20. Evers, Hans-Dieter and Anna-Katharina Hornidge (2007). Knowledge Hubs Along the Straits of Malacca.
- **21.** Sultana, Nayeem (2007). Trans-National Identities, Modes of Networking and Integration in a Multi-Cultural Society. A Study of Migrant Bangladeshis in Peninsular Malaysia.
- **22.** Yalcin, Resul and Peter M. Mollinga (2007). Institutional Transformation in Uzbekistan's Agricultural and Water Resources Administration: The Creation of a New Bureaucracy.
- 23. Menkhoff, T.; Loh, P. H. M.; Chua, S. B.; Evers, H.-D. and Chay Yue Wah (2007). Riau Vegetables for Singapore Consumers: A Collaborative Knowledge-Transfer Project Across the Straits of Malacca.
- 24. Evers, Hans-Dieter and Solvay Gerke (2007). Social and Cultural Dimensions of Market Expansion.
- **25.** Obeng, G. Y.; Evers, H.-D.; Akuffo, F. O., Braimah, I. and A. Brew-Hammond (2007). Solar PV Rural Electrification and Energy-Poverty Assessment in Ghana: A Principal Component Analysis.
- **26.** Eguavoen, Irit; E. Youkhana (2008). Small Towns Face Big Challenge. The Management of Piped Systems after the Water Sector Reform in Ghana.
- 27. Evers, Hans-Dieter (2008). Knowledge Hubs and Knowledge Clusters: Designing a Knowledge Architecture for Development

- **28.** Ampomah, Ben Y.; Adjei, B. and E. Youkhana (2008). The Transboundary Water Resources Management Regime of the Volta Basin.
- **29.** Saravanan.V.S.; McDonald, Geoffrey T. and Peter P. Mollinga (2008). Critical Review of Integrated Water Resources Management: Moving Beyond Polarised Discourse.
- **30.** Laube, Wolfram; Awo, Martha and Benjamin Schraven (2008). Erratic Rains and Erratic Markets: Environmental change, economic globalisation and the expansion of shallow groundwater irrigation in West Africa.
- 31. Mollinga, Peter P. (2008). For a Political Sociology of Water Resources Management.
- **32.** Hauck, Jennifer; Youkhana, Eva (2008). Histories of water and fisheries management in Northern Ghana.
- **33.** Mollinga, Peter P. (2008). The Rational Organisation of Dissent. Boundary concepts, boundary objects and boundary settings in the interdisciplinary study of natural resources management.
- 34. Evers, Hans-Dieter; Gerke, Solvay (2009). Strategic Group Analysis.
- **35.** Evers, Hans-Dieter; Benedikter, Simon (2009). Strategic Group Formation in the Mekong Delta The Development of a Modern Hydraulic Society.
- **36.** Obeng, George Yaw; Evers, Hans-Dieter (2009). Solar PV Rural Electrification and Energy-Poverty: A Review and Conceptual Framework With Reference to Ghana.
- **37.** Scholtes, Fabian (2009). Analysing and explaining power in a capability perspective.
- 38. Eguavoen, Irit (2009). The Acquisition of Water Storage Facilities in the Abay River Basin, Ethiopia.
- **39.** Hornidge, Anna-Katharina; Mehmood Ul Hassan; Mollinga, Peter P. (2009). 'Follow the Innovation' A joint experimentation and learning approach to transdisciplinary innovation research.
- 40. Scholtes, Fabian (2009). How does moral knowledge matter in development practice, and how can it be researched?
- **41.** Laube, Wolfram (2009). Creative Bureaucracy: Balancing power in irrigation administration in northern Ghana.
- 42. Laube, Wolfram (2009). Changing the Course of History? Implementing water reforms in Ghana and South Africa.
- **43.** Scholtes, Fabian (2009). Status quo and prospects of smallholders in the Brazilian sugarcane and ethanol sector: Lessons for development and poverty reduction.
- **44.** Evers, Hans-Dieter; Genschick, Sven; Schraven, Benjamin (2009). Constructing Epistemic Landscapes: Methods of GIS-Based Mapping.
- **45.** Saravanan V.S. (2009). Integration of Policies in Framing Water Management Problem: Analysing Policy Processes using a Bayesian Network.
- **46.** Saravanan V.S. (2009). Dancing to the Tune of Democracy: Agents Negotiating Power to Decentralise Water Management.
- **47.** Huu, Pham Cong; Rhlers, Eckart; Saravanan, V. Subramanian (2009). Dyke System Planing: Theory and Practice in Can Tho City, Vietnam.
- **48.** Evers, Hans-Dieter; Bauer, Tatjana (2009). Emerging Epistemic Landscapes: Knowledge Clusters in Ho Chi Minh City and the Mekong Delta.
- **49.** Reis, Nadine; Mollinga, Peter P. (2009). Microcredit for Rural Water Supply and Sanitation in the Mekong Delta. Policy implementation between the needs for clean water and 'beautiful latrines'.
- **50.** Gerke, Solvay; Ehlert, Judith (2009). Local Knowledge as Strategic Resource: Fishery in the Seasonal Floodplains of the Mekong Delta, Vietnam
- **51.** Schraven, Benjamin; Eguavoen, Irit; Manske, Günther (2009). Doctoral degrees for capacity development: Results from a survey among African BiGS-DR alumni.
- 52. Nguyen, Loan (2010). Legal Framework of the Water Sector in Vietnam.
- 53. Nguyen, Loan (2010). Problems of Law Enforcement in Vietnam. The Case of Wastewater Management in Can Tho City.
- **54.** Oberkircher, Lisa et al. (2010). Rethinking Water Management in Khorezm, Uzbekistan. Concepts and Recommendations.
- **55.** Waibel, Gabi (2010). State Management in Transition: Understanding Water Resources Management in Vietnam.
- **56.** Saravanan V.S.; Mollinga, Peter P. (2010). Water Pollution and Human Health. Transdisciplinary Research on Risk Governance in a Complex Society.
- **57.** Vormoor, Klaus (2010). Water Engineering, Agricultural Development and Socio-Economic Trends in the Mekong Delta, Vietnam.
- **58.** Hornidge, Anna-Katharina; Kurfürst, Sandra (2010). Envisioning the Future, Conceptualising Public Space. Hanoi and Singapore Negotiating Spaces for Negotiation.

- **59.** Mollinga, Peter P. (2010). Transdisciplinary Method for Water Pollution and Human Health Research.
- 60. Youkhana, Eva (2010). Gender and the development of handicraft production in rural Yucatán/Mexico.
- 61. Naz, Farha; Saravanan V. Subramanian (2010). Water Management across Space and Time in India.
- **62.** Evers, Hans-Dieter; Nordin, Ramli, Nienkemoer, Pamela (2010). Knowledge Cluster Formation in Peninsular Malaysia: The Emergence of an Epistemic Landscape.
- **63.** Mehmood UI Hassan; Hornidge, Anna-Katharina (2010). 'Follow the Innovation' The second year of a joint experimentation and learning approach to transdisciplinary research in Uzbekistan.
- 64. Mollinga, Peter P. (2010). Boundary concepts for interdisciplinary analysis of irrigation water management in South Asia.
- **65.** Noelle-Karimi, Christine (2006). Village Institutions in the Perception of National and International Actors in Afghanistan. (Amu Darya Project Working Paper No. 1)
- 66. Kuzmits, Bernd (2006). Cross-bordering Water Management in Central Asia. (Amu Darya Project Working Paper No. 2)
- **67.** Schetter, Conrad; Glassner, Rainer; Karokhail, Masood (2006). Understanding Local Violence. Security Arrangements in Kandahar, Kunduz and Paktia. (Amu Darya Project Working Paper No. 3)
- **68.** Shah, Usman (2007). Livelihoods in the Asqalan and Sufi-Qarayateem Canal Irrigation Systems in the Kunduz River Basin. (Amu Darya Project Working Paper No. 4)
- **69.** ter Steege, Bernie (2007). Infrastructure and Water Distribution in the Asqalan and Sufi-Qarayateem Canal Irrigation Systems in the Kunduz River Basin. (Amu Darya Project Working Paper No. 5)
- **70.** Mielke, Katja (2007). On The Concept of 'Village' in Northeastern Afghanistan. Explorations from Kunduz Province. (Amu Darya Project Working Paper No. 6)
- **71.** Mielke, Katja; Glassner, Rainer; Schetter, Conrad; Yarash, Nasratullah (2007). Local Governance in Warsaj and Farkhar Districts. (Amu Darya Project Working Paper No. 7)
- 72. Meininghaus, Esther (2007). Legal Pluralism in Afghanistan. (Amu Darya Project Working Paper No. 8)
- **73.** Yarash, Nasratullah; Smith, Paul; Mielke, Katja (2010). The fuel economy of mountain villages in Ishkamish and Burka (Northeast Afghanistan). Rural subsistence and urban marketing patterns. (Amu Darya Project Working Paper No. 9)
- 74. Oberkircher, Lisa (2011). 'Stay We Will Serve You Plov!'. Puzzles and pitfalls of water research in rural Uzbekistan.
- **75.** Shtaltovna, Anastasiya; Hornidge, Anna-Katharina; Mollinga, Peter P. (2011). The Reinvention of Agricultural Service Organisations in Uzbekistan a Machine-Tractor Park in the Khorezm Region.
- **76.** Stellmacher, Till; Grote, Ulrike (2011). Forest Coffee Certification in Ethiopia: Economic Boon or Ecological Bane?
- 77. Gatzweiler, Franz W.; Baumüller, Heike; Ladenburger, Christine; von Braun, Joachim (2011). Marginality. Addressing the roots causes of extreme poverty.
- **78.** Mielke, Katja; Schetter, Conrad; Wilde, Andreas (2011). Dimensions of Social Order: Empirical Fact, Analytical Framework and Boundary Concept.
- **79.** Yarash, Nasratullah; Mielke, Katja (2011). The Social Order of the Bazaar: Socio-economic embedding of Retail and Trade in Kunduz and Imam Sahib
- **80.** Baumüller, Heike; Ladenburger, Christine; von Braun, Joachim (2011). Innovative business approaches for the reduction of extreme poverty and marginality?
- 81. Ziai, Aram (2011). Some reflections on the concept of 'development'.
- 82. Saravanan V.S., Mollinga, Peter P. (2011). The Environment and Human Health An Agenda for Research.
- **83.** Eguavoen, Irit; Tesfai, Weyni (2011). Rebuilding livelihoods after dam-induced relocation in Koga, Blue Nile basin, Ethiopia.
- **84.** Eguavoen, I., Sisay Demeku Derib et al. (2011). Digging, damming or diverting? Small-scale irrigation in the Blue Nile basin, Ethiopia.
- 85. Genschick, Sven (2011). Pangasius at risk Governance in farming and processing, and the role of different capital.
- **86.** Quy-Hanh Nguyen, Hans-Dieter Evers (2011). Farmers as knowledge brokers: Analysing three cases from Vietnam's Mekong Delta.
- **87.** Poos, Wolf Henrik (2011). The local governance of social security in rural Surkhondarya, Uzbekistan. Post-Soviet community, state and social order.
- **88.** Graw, Valerie; Ladenburger, Christine (2012). Mapping Marginality Hotspots. Geographical Targeting for Poverty Reduction.
- 89. Gerke, Solvay; Evers, Hans-Dieter (2012). Looking East, looking West: Penang as a Knowledge Hub.

- **90.** Turaeva, Rano (2012). Innovation policies in Uzbekistan: Path taken by ZEFa project on innovations in the sphere of agriculture.
- 91. Gleisberg-Gerber, Katrin (2012). Livelihoods and land management in the Ioba Province in south-western Burkina Faso.
- **92.** Hiemenz, Ulrich (2012). The Politics of the Fight Against Food Price Volatility Where do we stand and where are we heading?
- **93.** Baumüller, Heike (2012). Facilitating agricultural technology adoption among the poor: The role of service delivery through mobile phones.
- **94.** Akpabio, Emmanuel M.; Saravanan V.S. (2012). Water Supply and Sanitation Practices in Nigeria: Applying Local Ecological Knowledge to Understand Complexity.
- **95.** Evers, Hans-Dieter; Nordin, Ramli (2012). The Symbolic Universe of Cyberjaya, Malaysia.
- **96.** Akpabio, Emmanuel M. (2012). Water Supply and Sanitation Services Sector in Nigeria: The Policy Trend and Practice Constraints.
- **97.** Boboyorov, Hafiz (2012). Masters and Networks of Knowledge Production and Transfer in the Cotton Sector of Southern Tajikistan.
- **98.** Van Assche, Kristof; Hornidge, Anna-Katharina (2012). Knowledge in rural transitions formal and informal underpinnings of land governance in Khorezm.
- **99.** Eguavoen, Irit (2012). Blessing and destruction. Climate change and trajectories of blame in Northern Ghana.
- **100.** Callo-Concha, Daniel; Gaiser, Thomas and Ewert, Frank (2012). Farming and cropping systems in the West African Sudanian Savanna. WASCAL research area: Northern Ghana, Southwest Burkina Faso and Northern Benin.
- **101.** Sow, Papa (2012). Uncertainties and conflicting environmental adaptation strategies in the region of the Pink Lake, Senegal.
- **102.** Tan, Siwei (2012). Reconsidering the Vietnamese development vision of "industrialisation and modernisation by 2020".
- 103. Ziai, Aram (2012). Postcolonial perspectives on 'development'.
- **104.** Kelboro, Girma; Stellmacher, Till (2012). Contesting the National Park theorem? Governance and land use in Nech Sar National Park, Ethiopia.
- **105.** Kotsila, Panagiota (2012). "Health is gold": Institutional structures and the realities of health access in the Mekong Delta, Vietnam.
- **106.** Mandler, Andreas (2013). Knowledge and Governance Arrangements in Agricultural Production: Negotiating Access to Arable Land in Zarafshan Valley, Tajikistan.
- **107.** Tsegai, Daniel; McBain, Florence; Tischbein, Bernhard (2013). Water, sanitation and hygiene: the missing link with agriculture.
- **108.** Pangaribowo, Evita Hanie; Gerber, Nicolas; Torero, Maximo (2013). Food and Nutrition Security Indicators: A Review.
- **109.** von Braun, Joachim; Gerber, Nicolas; Mirzabaev, Alisher; Nkonya Ephraim (2013). The Economics of Land Degradation.
- 110. Stellmacher, Till (2013). Local forest governance in Ethiopia: Between legal pluralism and livelihood realities.
- **111.** Evers, Hans-Dieter; Purwaningrum, Farah (2013). Japanese Automobile Conglomerates in Indonesia: Knowledge Transfer within an Industrial Cluster in the Jakarta Metropolitan Area.
- **112.** Waibel, Gabi; Benedikter, Simon (2013). The formation water user groups in a nexus of central directives and local administration in the Mekong Delta, Vietnam.
- **113.** Ayaribilla Akudugu, Jonas; Laube, Wolfram (2013). Implementing Local Economic Development in Ghana: Multiple Actors and Rationalities.
- **114.** Malek, Mohammad Abdul; Hossain, Md. Amzad; Saha, Ratnajit; Gatzweiler, Franz W. (2013). Mapping marginality hotspots and agricultural potentials in Bangladesh.
- **115.** Siriwardane, Rapti; Winands, Sarah (2013). Between hope and hype: Traditional knowledge(s) held by marginal communities.
- **116.** Nguyen Thi Phuong Loan (2013). The Legal Framework of Vietnam's Water Sector: Update 2013.
- **117.** Shtaltovna, Anastasiya (2013). Knowledge gaps and rural development in Tajikistan. Agricultural advisory services as a panacea?

### **ZEF Development Studies**

edited by Solvay Gerke and Hans-Dieter Evers

Center for Development Research (ZEF), University of Bonn

Shahjahan H. Bhuiyan Benefits of Social Capital. Urban Solid Waste Management in Bangladesh Vol. 1, 2005, 288 p., 19.90 EUR, br. ISBN 3-8258-8382-5

### Veronika Fuest

Demand-oriented Community Water Supply in Ghana. Policies, Practices and Outcomes Vol. 2, 2006, 160 p., 19.90 EUR, br. ISBN 3-8258-9669-2

Anna-Katharina Hornidge Knowledge Society. Vision and Social Construction of Reality in Germany and Singapore Vol. 3, 2007, 200 p., 19.90 EUR, br. ISBN 978-3-8258-0701-6

Wolfram Laube Changing Natural Resource Regimes in Northern Ghana. Actors, Structures and Institutions Vol. 4, 2007, 392 p., 34.90 EUR, br. ISBN 978-3-8258-0641-5

Lirong Liu Wirtschaftliche Freiheit und Wachstum. Eine international vergleichende Studie Vol. 5, 2007, 200 p., 19.90 EUR, br. ISBN 978-3-8258-0701-6

Phuc Xuan To Forest Property in the Vietnamese Uplands. An Ethnography of Forest Relations in Three Dao Villages Vol. 6, 2007, 296 p., 29.90 EUR, br. ISBN 978-3-8258-0773-3 Caleb R.L. Wall, Peter P. Mollinga (Eds.) Fieldwork in Difficult Environments. Methodology as Boundary Work in Development Research Vol. 7, 2008, 192 p., 19.90 EUR, br. ISBN 978-3-8258-1383-3

Solvay Gerke, Hans-Dieter Evers, Anna-K. Hornidge (Eds.) *The Straits of Malacca. Knowledge and Diversity* Vol. 8, 2008, 240 p., 29.90 EUR, br. ISBN 978-3-8258-1383-3

### Caleb Wall

Argorods of Western Uzbekistan. Knowledge Control and Agriculture in Khorezm Vol. 9, 2008, 384 p., 29.90 EUR, br. ISBN 978-3-8258-1426-7

### Irit Eguavoen

The Political Ecology of Household Water in Northern Ghana Vol. 10, 2008, 328 p., 34.90 EUR, br. ISBN 978-3-8258-1613-1

Charlotte van der Schaaf Institutional Change and Irrigation Management in Burkina Faso. Flowing Structures and Concrete Struggles Vol. 11, 2009, 344 p., 34.90 EUR, br. ISBN 978-3-8258-1624-7

Nayeem Sultana The Bangladeshi Diaspora in Peninsular Malaysia. Organizational Structure, Survival Strategies and Networks Vol. 12, 2009, 368 p., 34.90 EUR, br. ISBN 978-3-8258-1629-2

Peter P. Mollinga, Anjali Bhat, Saravanan V.S. (Eds.) When Policy Meets Reality. Political Dynamics and the Practice of Integration in Water Resources Management Reform Vol. 13, 2010, 216 p., 29.90 EUR, br., ISBN 978-3-643-10672-8 Irit Eguavoen, Wolfram Laube (Eds.) Negotiating Local Governance. Natural Resources Management at the Interface of Communities and the State Vol. 14, 2010, 248 p., 29.90 EUR, br., ISBN 978-3-643-10673-5

William Tsuma Gold Mining in Ghana. Actors, Alliances and Power Vol. 15, 2010, 256 p., 29.90 EUR, br., ISBN 978-3-643-10811-1

### Thim Ly

Planning the Lower Mekong Basin: Social Intervention in the Se San River Vol. 16, 2010, 240 p., 29.90 EUR, br., ISBN 978-3-643-10834-0

### Tatjana Bauer

The Challenge of Knowledge Sharing - Practices of the Vietnamese Science Community in Ho Chi Minh City and the Mekong Delta Vol. 17, 2011, 304 p., 29.90 EUR, br., ISBN 978-3-643-90121-7

Pham Cong Huu Floods and Farmers - Politics, Economics and Environmental Impacts of Dyke Construction in the Mekong Delta / Vietnam Vol. 18, 2012, 200 p., 29.90 EUR, br., ISBN 978-3-643-90167-5

Judith Ehlert Beautiful Floods - Environmental Knowledge and Agrarian Change in the Mekong Delta, Vietnam Vol. 19, 2012, 256 S., 29,90 EUR, br, ISBN 978-3-643-90195-8

Nadine Reis Tracing and Making the State - Policy practices and domestic water supply in the Mekong Delta, Vietnam Vol. 20, 2012, 272 S., 29.90 EUR, br., ISBN 978-3-643-90196-5 Martha A. Awo Marketing and Market Queens - A study of tomato farmers in the Upper East region of Ghana Vol. 21, 2012, 192 S., 29.90 EUR, br., ISBN 978-3-643-90234-4

### Asghar Tahmasebi

Pastoral Vulnerability to Socio-political and Climate Stresses - The Shahsevan of North Iran Vol. 22, 2013, 192 S., 29.90 EUR, br., ISBN 978-3-643-90357-0

### Anastasiya Shtaltovna

Servicing Transformation - Agricultural Service Organisations and Agrarian Change in Post-Soviet Uzbekistan Vol. 23, 2013, 216 S., 29.90 EUR, br., ISBN 978-3-643-90358-7

### Hafiz Boboyorov

Collective Identities and Patronage Networks in Southern Tajikistan Vol. 24, 2013, 304 S., 34.90 EUR, br., ISBN 978-3-643-90382-2



Zentrum für Entwicklungsforschung Center for Development Research University of Bonn

# **Working Paper Series**

Authors:	Anastasiya Shtaltovna (ZEF)
Contact:	shtaltov@uni-bonn.de
Photo:	Anastasiya Shtaltovna (ZEF)

Published by: Zentrum für Entwicklungsforschung (ZEF) Center for Development Research Walter-Flex-Straße 3 D – 53113 Bonn Germany

Phone: +49-228-73-1861 Fax: +49-228-73-1869 E-Mail: zef@uni-bonn.de

www.zef.de